



Article The mPOC Framework: An Autonomous Outbreak Prediction and Monitoring Platform Based on Wearable IoMT Approach

Sasan Adibi D

School of Information Technology, Deakin University, Geelong, VIC 3220, Australia; sasan.adibi@deakin.edu.au

Abstract: This paper presents the mHealth Predictive Outbreak for COVID-19 (mPOC) framework, an autonomous platform based on wearable Internet of Medical Things (IoMT) devices for outbreak prediction and monitoring. It utilizes real-time physiological and environmental data to assess user risk. The framework incorporates the analysis of psychological and user-centric data, adopting a combination of top-down and bottom-up approaches. The mPOC mechanism utilizes the bidirectional Mobile Health (mHealth) Disaster Recovery System (mDRS) and employs an intelligent algorithm to calculate the Predictive Exposure Index (PEI) and Deterioration Risk Index (DRI). These indices trigger warnings to users based on adaptive threshold criteria and provide updates to the Outbreak Tracking Center (OTC). This paper provides a comprehensive description and analysis of the framework's mechanisms and algorithms, complemented by the performance accuracy evaluation. By leveraging wearable IoMT devices, the mPOC framework showcases its potential in disease prevention and control during pandemics, offering timely alerts and vital information to healthcare professionals and individuals to mitigate outbreaks' impact.

Keywords: Mobile Health (mHealth); COVID-19; Biological Watch (Bio-Watch); mHealth Predictive Outbreak (mPOC); Predictive Exposure Index (PEI); Deterioration Risk Index (DRI); 5th Generation Mobile Network (5G); Internet of Medical Things (IoMT); Disaster Recovery System (mDRS)



Citation: Adibi, S. The mPOC Framework: An Autonomous Outbreak Prediction and Monitoring Platform Based on Wearable IoMT Approach. *Future Internet* **2023**, *15*, 257. https://doi.org/10.3390/ fi15080257

Academic Editor: Matthew Pediaditis

Received: 3 July 2023 Revised: 22 July 2023 Accepted: 27 July 2023 Published: 30 July 2023



Copyright: © 2023 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

1. Introduction

The current COVID-19 pandemic has presented significant worldwide challenges, with millions of cases and fatalities reported to date [1]. In response, governments and global health organizations have emphasized the importance of vaccination, social distancing, and mask usage to mitigate the spread of the virus [2]. Airborne transmission, particularly in indoor spaces, poses a higher risk of viral exposure, leading to more severe illness and extended hospital stays [3]. Timely diagnosis and effective management are crucial for increasing survival rates, including early detection during asymptomatic stages and proper containment measures [4].

Besides the COVID-19 pandemic, infectious disease outbreaks pose significant challenges to public health systems worldwide, which can lead to increased morbidity and mortality. The recent COVID-19 pandemic highlighted the critical need for effective outbreak management and disease prevention strategies. Traditional methods of outbreak management often relied on retrospective data analysis and manual reporting, leading to delays in detecting and responding to emerging threats. To address these challenges, there is an increasing demand for innovative and data-driven approaches that can provide timely and accurate outbreak predictions. Therefore, because the traditional surveillance methods may not be agile enough to capture real-time data, leading to delays in recognizing the emergence of new infections, the implementation of necessary interventions, timely detection, and containment of outbreaks are essential to minimize the impact on public health.

Furthermore, the dynamic nature of infectious diseases requires continuous monitoring and surveillance to track their evolution and spread. Wearable Internet of Medical Things (IoMT) technologies offer a potential solution to address this challenge. These devices, such as smartwatches, wristbands, and other mobile-based gadgets, provide a means for continuous monitoring of individuals' physiological and environmental data. They can capture vital signs, body temperature, respiratory patterns, and other relevant health metrics in real-time, offering a wealth of information for outbreak prediction and monitoring.

By leveraging the continuous data collection capabilities of wearable devices, public health authorities and healthcare professionals can gain valuable insights into population health trends and detect early signs of disease outbreaks. This continuous monitoring enables proactive measures, such as implementing targeted interventions, resource allocation, and contact tracing, to prevent further spread of infectious diseases.

The critical importance of accurate outbreak predictions cannot be overstated. Early detection allows for timely response measures, including quarantine protocols, travel restrictions, and vaccination campaigns. By deploying an autonomous outbreak prediction and monitoring platform, such as the framework proposed in this paper, public health officials can optimize resource allocation, identify at-risk populations, and implement evidence-based measures to control the spread of infectious diseases.

Therefore, the challenges in outbreak management and disease prevention necessitate a paradigm shift towards data-driven approaches that prioritize timely and accurate outbreak predictions. Wearable IoMT technologies play a pivotal role in this endeavor, as they offer real-time data collection and continuous monitoring, enabling public health authorities and healthcare professionals to take proactive measures that reduce the impact of infectious diseases on public health. To that effect, the proposed mPOC: Mobile Health (mHealth) Predictive Outbreak for COVID-19 framework offers a promising solution to address these challenges by leveraging wearable devices and integrating user-centric and public health data to provide an innovative and effective tool for outbreak prediction and monitoring.

This paper introduces the mPOC framework, which is an extension to the mDRS (mHealth Disaster Recovery System), previously presented in reference [5]. While the mDRS focused on disaster recovery and post-disaster health management, it primarily addressed immediate response and recovery measures after a disaster or outbreak had already occurred. The mDRS played a crucial role in providing health services and support during emergencies, but its capabilities were limited to addressing the aftermath of an outbreak.

In contrast, the mPOC system was introduced to take outbreak management and disease prevention to the next level. It extends beyond disaster recovery to provide realtime outbreak prediction, continuous monitoring, and early warning capabilities. By leveraging the Internet of Medical Things (IoMT) and wearable devices, the mPOC system captures user-centric physiological data to gain valuable insights into an individual's health status and response to outbreak situations. The mPOC mechanism is designed to address the need for timely diagnosis and effective management of pandemic-related illnesses on a public health scale. It incorporates two distinctive and collaborative approaches, namely, a bottom-up and top-down approaches.

- (1) Bottom-up approach: This involved capturing user-centric physiological parameters (that will be discussed in Section 3), which offers valuable insights into an individual's well-being and response to outbreaks. Integrating these parameters into the mPOC system enables a comprehensive understanding of individuals' health status, leading to the evaluation of the Deterioration Risk Index (DRI), which measures users' medical risk levels. Concrete examples of data collection scenarios can facilitate personalized and timely outbreak predictions, thereby optimizing public health responses and ensuring individual well-being during infectious disease outbreaks. A bottom-up approach is used to collect and analyze user-centric data, including psychological factors, to assess and monitor individuals.
- (2) Top-down approach: Simultaneously, the mPOC's top-down approach involves transmission of information to and from the Outbreak Tracking Centre (OTC) and other public-health-related resources. The top-down approach complements the user-centric

data obtained in the bottom-up approach, providing valuable insights from a broader perspective. Examples of the top-down data collection include:

Public Health Data—The algorithm receives real-time data on confirmed cases, outbreak trends, and demographic information from health authorities and healthcare institutions. Analyzing this data helps assess outbreak severity, identify high-risk regions, and allocate resources effectively.

OTC Information—The mPOC framework establishes a bi-directional connection with the OTC to transmit daily updates on outbreak progression, containment strategies, and healthcare capacities. This enables timely adjustments to predictions and response strategies.

Environmental Data—Integrating data from environmental sensors and weather databases provide insights into factors influencing disease transmission. For instance, monitoring air pollution, wind speed, and humidity levels aids in identifying potential hotspots for rapid outbreak spread.

Healthcare Infrastructure and Resource Availability—Data on hospital bed capacities and medical supplies contributes to effective outbreak response planning. Assessing available resources helps determine the healthcare system's readiness and facilitates resource allocation.

Mobility and Transportation Data—Analyzing data from transportation systems, such as public transit usage and travel patterns, offers insights into population movement and its impact on disease spread. This data supports the optimization of containment measures, such as travel restrictions and quarantine protocols.

The mPOC algorithm evaluates the top-down data and calculates the Predictive Exposure Index (PEI). The values of DRI and PEI are then used to assess the current status and expected pandemic-related exposure, which help create appropriate warnings that are triggered using an adaptive threshold-based criteria. These warnings are then sent to the Outbreak Tracking Center (OTC) on a regular basis.

This paper provides a detailed analysis of the mechanisms and algorithm employed. To summarize, the main contributions of this paper are as follows:

- (1) The description of the mPOC framework, based on its intelligent algorithm;
- (2) Combining the bottom-up and top-down approaches, leading to the calculations of the DRI and PEI values from end-user and OTC information gatherings;
- (3) Leveraging wearable IoMT devices, based on the Bio-Watch concept;
- (4) By combining wearable devices, intelligent algorithms, and real-time data analysis, the system contributes to enhancing disease prevention and control efforts, providing valuable insights and timely alerts to mitigate the impact of outbreaks.

The structure of this paper is as follows: Section 2 presents a literature review on the technological interventions for COVID-19 monitoring. Section 3 explores non-invasive clinical and physiological presentations (CPPs) of COVID-19, discussing their specificity and sensitivity measures. Section 4 provides an overview of epidemiology and public health perspectives of COVID-19. Section 5 focuses on the sensory systems utilized in the mPOC framework. Section 6 delves into the mPOC algorithm and IoMT protocols. The feasibility, scalability, and performance evaluation of mPOC are discussed in Section 7. Section 8 outlines future directions for technology-driven pandemic response systems, followed by the conclusion and references.

In this paper we aim to address and respond to the following two research questions.

- 1. How can the mPOC framework effectively utilize wearable and mobile devices to autonomously predict and monitor outbreaks in pandemic-stricken areas?
- 2. What are the key features and functionalities of the mPOC system that make it superior to existing technology-assisted intervention systems for monitoring COVID-19 and preventing the spread of the virus?

These research questions focus on understanding the effectiveness and unique aspects of the mPOC framework in outbreak prediction and monitoring, as well as its advantages over other technology-assisted interventions.

2. Literature Review on the Technological Interventions for COVID-19 Monitoring

The COVID-19 pandemic has brought about significant and long-lasting changes in the healthcare system and has emphasized the need for effective technology-assisted interventions. In this section, we present a comprehensive review of current intervention systems, specifically focusing on the utilization of mobile-based applications. A total of 30 recent studies conducted between 2020 and 2023 were selected and analyzed to examine the various smartphone applications used for monitoring the COVID-19 pandemic and its associated symptoms. These applications were primarily employed for active and passive screening, as well as contact tracing [6,7].

2.1. Study Selection Criteria

The 30 studies included in this review (references [4–18]) were carefully selected based on the following criteria:

- Mobile Health (mHealth) Devices: The focus of these studies on the utilization of app-based digital devices capable of collecting and/or analyzing physiological information were considered. This includes smartphones, smartwatches, tablets, and wearable sensors.
- Human Subjects: In each of the studies, a minimum of 10 subjects are included to ensure an adequate sample size for reliable conclusions.
- Health Monitoring: These studies were specifically designed for monitoring COVID-19 signs and symptoms and ensuring relevance to the recent COVID-19 pandemic.
- Quality Publications: Only peer-reviewed publications were included in this review, ensuring the reliability and credibility of the research.
- Year of Publication: The considered studies in the literature review were published between January 2020 and June 2023, encompassing a wide range of coverage in technology-assisted interventions for the last three years in COVID-19 monitoring.

By applying these rigorous selection criteria, we aimed to ensure the inclusion of highquality studies that provide valuable insights into the use of mobile devices for monitoring and managing the COVID-19 pandemic.

2.2. Literature Review Assessments

A total of 30 research studies were reviewed, encompassing 1047 mobile apps. Among these studies, 83% (25 out of 30 studies) were conducted in developed countries, with the United States of America accounting for 40% (10 out of 25 studies). Other countries involved in the studies included the United Kingdom (UK), Germany, Spain, Denmark, France, Sweden, Norway, Switzerland, Australia, Korea, Hong Kong, India, Taiwan, and Japan. The participant demographics in these studies were diverse, covering variations in genders, age groups, ethnicities, professions, and health conditions, as discussed in Section 2.1.

The participants in the studies were from various backgrounds, including:

COVID-19 patients (87%) ([4,6–13,15–18]), global participants (67%) (references [4, 6,7,9,10,12,14,15,17,18]), healthcare workers (27%) ([4,12,13,17,18]), academics (staffs and students) (20%) (references [4,11,16]), elderly (13%) ([4,9], [14,16,18]), and adults with specific medical conditions (20%) ([8,9,13]). Furthermore, the mHealth tools and techniques utilized in these studies varied. Approximately 73% of the studies used mobile (iOS or Android) apps, while 3% utilized tablets and SMS. Additionally, 40% of the studies integrated wristbands or smartwatches with necessary apps. Table 1 summarizes the findings from the research studies.

				Participants			
Kef. No.	Country	Age Group	Occupations	Performance Metric	Bottom Up	Top Down	Research Method/Tools ¹
[4]	Global	18-60+	Healthcare	Availability	1	1	App, SW, WB, OMG, Web
[5]	Australia	18+	Diverse	MOS-R ²	1	1	App, SW, WB, OMG, Web
[6]	UK	>40	Diverse (Patients)	App uptake measures	✓	1	App, SW, WB, OMG, Web
[7]	Global	18+	Diverse (Patients)	Questionnaire	1	Х	Арр
[8]	Iran	18+	Diverse (Patients)	Questionnaire	1	Х	App, web
[9]	Global	52–74	Diverse (Patients)	Feasibility, reliability	1	Х	Web (iPad)
[10]	Global	18+	Healthcare	Delay	✓	Х	App, web
[11]	India	18+	Academics	Technology Acceptance	1	1	App, SW, OMG, Web
[12]	Global	19+	Healthcare	Data analysis	1	1	App, OMG, Web
[13]	USA	16–61	Healthcare	Surveys	1	1	App, OMG, Web
[14]	Germany	45-83	Healthcare	User Experience survey	1	Х	App, OMG, Web
[15]	Singapore	15–50+	Diverse (Patients)	Technology adoption	1	1	App, web
[16]	Switzerland	18–79	Academics	Technology Acceptance	1	Х	App, Web
[17]	Germany	18-60+	Healthcare	Usability	1	Х	App, OMG, Web
[18]	Pakistan	18-80	Healthcare	Availability	1	Х	App, SW, WB, OMG, Web

Table 1.	Summary	of the	literature	review	findings.
----------	---------	--------	------------	--------	-----------

¹ Mobile applications: App; Smartwatch: SW; Wristband: WB; Web applications: Web; Other Mobile-based gadgets: OMG; ² Mean Opinion Score: MOS; Rating Factor: R (quality of service measure in VoIP systems). \checkmark : supported; X: not supported.

Table 1 presents the research results that focused on various participants from global populations aged 15 to over 80 years of age with diverse backgrounds, including healthcare professionals, patients, academics, and individuals from different countries, including Australia, Iran, India, USA, Germany, Singapore, Switzerland, and Pakistan. The performance metrics evaluated in the studies varied, including system availability, app uptake measures, feasibility, reliability, delay measures, technology acceptance, health data analysis, user experience, technology adoption, and system usability. The majority of the studies were based on bottom-up approaches, with a few including top-down approaches. The research method and tools used for data collection and analysis included: mobile applications, smartwatches, wristbands, web applications, and other mobile-based gadgets.

In contrast to these studies, the proposed mPOC solution offers several unique features that set it apart from the systems and devices used in the reviewed studies. Firstly, it leverages the combination of a smartwatch and a smartphone app, with potential extensions to tablets and laptops/PCs without direct user interventions (apart from the initialization step) in a non-invasive approach. Secondly, the mPOC solution is designed to easily scale up to the population level. Thirdly, it is adaptive, meaning it can adapt to an individual user's physiological information, leading to higher accuracy compared to traditional systems and approaches. These distinguishing features will be discussed in detail in the subsequent sections.

3. Non-Invasive Clinical and Physiological Presentations (CPPs) of COVID-19

The impact of diseases, such as cancers, often involves clinical diagnoses that require invasive procedures such as biopsies and pathological tests. In contrast, non-invasive approaches are preferred by patients as they allow medical procedures to be conducted more comfortably, avoiding the inconvenience and pain associated with invasive procedures. However, these technology-based methods must receive approval from government and regulatory agencies, such as the Food and Drug Administration (FDA), before complete autonomous usage is permitted. Once approved, these methods can be easily scaled up for public utilization, alleviating the burden on overwhelmed healthcare systems, as experienced by many countries during pandemics [18].

This article introduces a novel approach for monitoring COVID-19 symptoms on a public health scale using the mPOC mechanism, which is based on the bidirectional mHealth Disaster Recovery System (mDRS) [5]. The mPOC system relies on the physiological symptoms and clinical presentations, which include fever, dry cough, sore throat, fatigue, and shortness of breath, which can be seen in severe cases and in elderly patients. The gastrointestinal symptoms (vomiting and diarrhea), cardiac presentations (myocarditis, pericarditis and/or atrial fibrillation), and loss/reduction of senses (smell and taste) are less commonly observed in recent COVID-19 subvariants (e.g., XBB) [19,20]. By utilizing these clinical and physiological presentations (CPPs) within its algorithm, the mPOC system assesses the user's health condition and calculates two crucial indices: the Predictive Exposure Index (PEI) and the Deterioration Risk Index (DRI). These indices trigger appropriate warnings to the user and facilitate the transmission of updates to and from the Outbreak Tracking Center (OTC). The selection and weighting of CPPs are determined by their non-invasive detectability, prevalence, specificity, sensitivity (accuracy), and demographicspecific considerations, enabling the system to adapt to individual users.

Before delving into the list of CPPs, it is important to define several terminologies associated with the characteristics of CPPs:

Prevalence (PR): Percentage of diagnosed (or to be diagnosed) people displaying the physiological presentation.

Timeline: The entire timeline of the disease can be divided into several segments:

Incubation Period (IP)—The period right after the initial exposure to the virus until the time that the initial symptoms appear, which can take up to 5 days for XBB Omicron subvariant (XBB 1.15 and 1.16) [19].

Early Stage (ES)—The duration of the initial signs and symptoms of Omicron subvariants, which can take anywhere from 1 to 5 days, with an average of 3.63 days (95%CI: 3.25–4.02 days). During this stage the patient exhibits flu-like symptoms [21].

Advanced Stage (AS)—This is when the disease progresses into a more serious stage and the patient's overall health starts to decline more rapidly. This could happen between 5 to 7 days after the initial exposure. The final stage follows the AS, which can either lead to the Recovery Stage (RS), when the patient starts to feel better and recovers from the disease (at least 3 days after last main CPPs), or it progresses into a Terminal Stage (TS), when the patient enters a critical phase, often requiring a ventilator or a breathing tube. After TS, the patient either survives and enters the RS or passes away.

Specificity and Sensitivity (SS): Is the percentage of accuracy of diagnosis based on the detection of a specific CPP. Overall, diagnosing COVID-19 is linked to a relatively low sensitivity and a high specificity, meaning the initial home-based tests (RAT: Rapid Antigen Test) may initially yield negative results while the patient has COVID-19 and, eventually after a few tests, the result will turn positive. However, if the test result becomes positive, it is with a high percentage (over 95%) of certainty that the patient is actually COVID positive.

Demographic Factor (DF): The algorithm is initialized based on initial values of prevalence, timelines, and specificity and sensitivity of each CPP and once the algorithm learns about the user's demographic data (e.g., age, gender, ethnicity, BMI (Body Mass Index), preexisting conditions, etc.), the more accurate weights for the CPP will be calculated. The capabilities of the current smartphones and the feasibility of utilizing body-worn sensors enable the non-invasive detection of the following CPPs: dry cough, fever, heartbeat rate, fatigue (manifested as unusual inactivity), shortness of breath (dyspnea), nasal congestion, confusion (manifested in abnormal EEG "electroencephalography" readings), and low blood oxygen level (hypoxemia). Pre-existing medical conditions can also be tracked by using body-worn sensors, such as electrocardiography (ECG), to track cardiovascular issues, and terahertz technology, to track diabetes [22].

In the following section, the characteristics of the early CPPs are discussed [23].

3.1. CPPs during Incubation Period (IP)

During the incubation period (IP), most patients typically show little to no signs; however, a subset of individuals may experience mild symptoms, such as fatigue and a low-grade fever. These subtle changes in body temperature and behavior can be detected by sensors.

3.2. CPPs during Early Stage (ES)

The early CPPs, which can be detected using a smartphone, have the following features and characteristic [24]:

Timeline: Early Stage (ES).

Prevalence: fever (67%), cough (42%), runny nose (rhinorrhea) (33.7%), body ache (14.5%), fatigue (14.1%), and shortness of breath (dyspnea) (9.6%) ([19-21]).

Specificity and Sensitivity: Each of these early CPPs individually has a sensitivity and specificity of around 50% for diagnosing COVID-19 using the Rapid Antigen Test (RAT). However, when combined with the Polymerase Chain Reaction (PCR) test, their accuracy can reach 95% [25].

This study focuses on tracking and identifying individuals during the incubation period (IP) and early stage (ES) of COVID-19, wherein they may exhibit minimal or no symptoms. It is assumed that individuals who are positive for COVID-19 would have already received treatment during the advanced stage (AS), recovery stage (RS), or terminal stage (TS). The IP period is considered crucial as it is a time when the infection is more likely to spread unknowingly, given that the median duration of asymptomatic infection is 2 days [25]. The proposed mPOC system aims to detect and monitor individuals during this critical phase.

4. Public Health Epidemiology of COVID-19

COVID-19, also known as SARS-CoV-2, belongs to the human coronavirus family of viruses, sharing genome similarities with SARS (Severe Acute Respiratory Syndrome, SARS-CoV) and MERS (Middle East Respiratory Syndrome, MERS-CoV). The transmission of SARS-CoV-2 occurs through droplet transmission (when droplets are >5–10 μ m in size), contact transmission (physical contact with contaminated surfaces and objects, followed by touching the mouth, nose, or eyes), and aerosol transmission (cloud of droplet particles <5 μ m in size). Both droplet and aerosol transmissions occur when an infected person coughs or sneezes and an uninfected person inhales the droplets. The virus can remain active on various surfaces for several days, and the contaminated droplets can stay airborne for up to an hour, particularly in indoor spaces [26]. Consequently, the risk of transmission increases in enclosed environments, such as shopping malls, restaurants, and cafes.

Public health agencies play a crucial role in providing the population with essential pandemic-related information. However, many individuals may struggle to understand and interpret the overwhelming volume of released and sometimes conflicting information. Therefore, the presence of an intelligent, user-centric device capable of interpreting and aligning this information with the user's demographic details becomes vital.

A fundamental epidemiological concept, known as the reproduction number, or R_0 , is utilized to assess the contagiousness of COVID-19. R_0 represents the average number of people that can be infected by a single sick individual. An R_0 value of 1 suggests that

number of infections. An R_0 value less than 1 indicates that an infected person, on average, transmits the virus to fewer than one person, leading to a gradual decline in the disease's spread. Conversely, an R_0 value greater than 1 signifies that an infected person, on average, transmits the virus to more than one person, resulting in a rapid spread of the disease, depending on the magnitude of R_0 . Initially, the estimated R_0 for COVID-19 ranged from 2.2 to 3.58 [27]. However, its contagiousness had increased, with R_0 values estimated to have been around 5.08 (95% CI 3.8–8.9) for the Delta variant, and with a 3-to-6-fold increase to 10–18.6 for the Omicron variant (BA.7 and XBB) (references [28–30]). In comparison, the flu has an R_0 of 1.4–1.6 [31,32].

The mPOC system utilizes the R_0 value to calculate the Predictive Exposure Index (PEI) and Deterioration Risk Index (DRI), which measure the likelihood of users infecting others based on sensed CPPs and other demographic factors. A higher PEI value indicates a higher likelihood of users exposing others to the virus, while a higher DRI value indicates an increased risk of the individual's health deterioration once exposure occurs.

5. mPOC Sensory System

In this section, we will discuss the design of the innovative mPOC system and its algorithm, which is based on the Internet of Medical Things (IoMT) protocol. As previously mentioned, mPOC has been developed as an extension of an existing solution called the mHealth Disaster Recovery System (mDRS) [5]. The foundation of both, mPOC and mDRS revolves around the utilization of the Bio-Watch, which captures physiological signals from users in conjunction with a smartphone running the mPOC/mDRS apps—which are used in disaster management. The capabilities of the mPOC system have been expanded beyond the conventional mDRS to cater to COVID-19 applications, including the three stages of the mPOC algorithm, the DRI and PEI evaluations, as well as the intelligent warning system that was not included in the mDRS. In the following section, the details of the mPOC system are presented.

5.1. Smartphone-Embedded Sensory Functionalities

Nowadays, smartphones are equipped with a variety of embedded (on-device) functionalities that serve various purposes. For instance, the microphone is essential for establishing bidirectional voice communication with humanitarian operators. Moreover, microphones can actively capture various physiological sounds, including the user's breathing and respiratory sounds, as well as ambient voices in the background.

The combination of a loudspeaker and visual signals is utilized to issue appropriate warnings to the user, when necessary, from the humanitarian operators.

The accelerometer and gyroscope sensors are capable of detecting acceleration/ deceleration along different axes using angular velocities. These sensors enable tracking of tremors, chills, and fatigue, which are characterized by relatively low mobility compared to the user's typical behavior profile. Additionally, sudden falls can also be detected by these sensors.

Furthermore, smartphones are equipped with a compass, as well as GPS and indoor tracking tools, allowing for the tracking of the user's location, heading, and direction in relation to reference points. This functionality proves helpful in monitoring and keeping track of the user's whereabouts.

5.2. The Physiological Watch (Bio-Watch)

The innovative concept of the Bio-Watch framework was introduced in reference [5]. The current article further incorporates various original and additional physiological parameters to enable autonomous outbreak prediction and monitoring. These parameters include brainwaves (electroencephalography; EEG), blood pressure, heart pulse rate (electrocardiography; ECG), body temperature, blood oxygen level (SpO2), muscle contractions (electromyography; EMG), and blood glucose levels (utilizing Terahertz technology [22]). Notably, the experimental results from testing ECG, SpO2, EEG, and EMG highlight the successful processing of sensory data for highly mobile systems [33].

Here we delve into the pertinent technologies deployed in the mPOC framework.

5.3. Fifth Generation (5G) Mobile Networks

The 4G technology, also known as Long Term Evolution (LTE), which was supported by the 3rd Generation Partnership Project (3GPP), had been a widely adopted mobile communication standard since 2010. The 5G technology was introduced in 2019 and while 5G and 4G LTE can coexist, 5G operates on various frequency spectrums, including below 1 GHz, sub 6 GHz, and 26–40 GHz (sub-mmWave). While the number of LTE/4G mobile subscriptions worldwide is expected to peak in 2023 at around 5.4 billion subscriptions and then gradually decline, the number of 5G subscribers continues to grow, and is forecasted to pass 1.5 billion in 2023 and reach 4.6 billion globally by the end of 2028 [34].

It should be noted that the sixth Generation (6G) mobile system standard is currently under development and is planned to be rolled out by 2030 [34].

5.4. Device-to-Device (D2D) Communications

Device-to-Device (D2D) supports the connectivity between mobile applications and is transparent to users. D2D is supported by both 4G-LTE (initiated with Release 12) and 5G. The current 3GPP (5G) is based on Release 18, which supports D2D communications [35–37]. In the context of the mPOC system, smartphones in the wireless range of one another can discover each other and exchange users' (e.g., health-related) information, in addition to the information received from the public health agencies, which are then used as a predictive measure to assess the dynamics of the moving population and forecast the possibility for further spread of the virus in a pandemic situation. This can be assisted by a mechanism provided by the LTE protocol, called LTE Direct (LTE-D), which uses line-of-sight communication, and it has a range of up to 500 m. Figure 1 shows the LTE-Direct one-to-one discovery and registration handshakes (adapted from [5]). The mPOC algorithm utilizes D2D, which will be discussed in the next section.

It's important to acknowledge that D2D can also be leveraged through other communication protocols, such as Bluetooth and NFC (Near Field Communication), which are out of the scope of this paper.



Figure 1. LTE-Direct D2D handshakes in mPOC system [5].

6. Novel mHealth Predictive Outbreak for COVID-19 (mPOC) Algorithm and IoMT Protocol

In this section, we present the design of our innovative mHealth Predictive Outbreak system for COVID-19 (mPOC). The successful operation of the mPOC system relies on the presence of an mPOC-ready application installed on users' smartphones. This application seamlessly integrates the functionalities of the smartphone, Bio-Watch, and public health information, as depicted in Figures 1 and 2.



Figure 2. The mHealth Pandemic Outbreak COVID-19 (mPOC) System [5].

The mPOC system incorporates an algorithm that progresses through three distinct stages, as illustrated in Figure 3. These stages will be comprehensively defined and discussed in the subsequent parts of this section.

6.1. Initial Stage

Once the mPOC application is successfully installed and launched on the smartphone, the algorithm initiates and collects user-related information, such as the user's location, including the suburb, city, and country. This location data can be obtained through various means, such as GPS, Wi-Fi, or mobile networks. Additionally, the application prompts the user to provide essential demographic details, including age, gender, ethnicity, height, weight, and any pre-existing health conditions that may influence the risk of COVID-19 complications; examples of such conditions encompass cancer, immunocompromised disorders, diabetes, pulmonary, renal, and cardiovascular diseases.

One notable complication that can arise from COVID-19 infections is the hyperactive immune response known as Cytokine Storm (CS) [38], wherein the immune system inundates the infection site (e.g., lungs) with inflammatory white blood cells, potentially leading to multiple organ failure. By conducting a blood test to assess the cytokine/chemokine profiles, it becomes possible to identify risk factors associated with Cytokine Storm, which can be considered a pre-existing condition. The DRI value, which indicates the health risk, is updated accordingly to reflect this information.



Figure 3. mPOC Protocol Flow-Chart.

Concurrent with the mPOC algorithm's execution on the smartphone, the Bio-Watch device attempts to establish a connection with the smartphone, calibrate its sensors, commence capturing the user's physiological indicators, and adapt to the typical readings. This stage occurs once and can be subsequently updated, as depicted in Table 2, to ensure accurate and reliable data captured for the mPOC system.

Table 2. The three stages of mPOC algorithm.

Stage	Device	Sensing/Information Input	Monitored CPPs/Health Conditions User Inputs			
Initial	Smartphone	Location (suburb, city, country), BMI, pre-existing conditions, job, number of members of household	Input: age, gender, ethnicity, height, weight Input: preexisting conditions: cancer, organ implant, hypertension, diabetes, and pulmonary/renal/cardiovascular diseases. Calculate: BMI/Cytokine Storm/age/gender risk factors→DRI			
	Bio-Watch	Sensory data input from smartphone	Calibrating sensors and learning user physiology			
IoMT	Smartphone	Microphone (voice/sound), camera (video/light), gyroscope, accelerometer, compass, GPS	Dry cough: Mic Nasal congestion: Mic	Disease monitoring: Restlessness/fatigue/insomnia: PSG		
	Bio-Watch	Temperature, accelerometer and gyroscope, compass, ECG, EMG, EEG, SpO2, BVP, SCL, THz, PSG (EEG, EMG, ECG, accelerometer, gyroscope, gyroscope)	Shortness of breath: SpO2 Hypertension: BVP Pain: BVP, ECG, SCL Fever: Temperature Tachycardia: ECG	Confusion: EEG Diabetes: THz Heart condition: ECG Diarrhea/vomit: Mic Data sent to PRaaS to calculate PEI		
PRaaS	Smartphone	LTE-Direct (D2D), PHBA, OTC	Anonymous exchange of health and PEI and DRI values are upo	onymous exchange of health details with users in wireless range d PEI and DRI values are updated and sent to OTC		
	Bio-Watch	Information exchange with smartphone	Information exchange with smartphone			

6.2. Internet of Medical Things (IoMT) Stage

The Internet of Medical Things (IoMT), in a classical sense, represents an interconnected network of medical devices, sensors, and software applications designed to collect, transmit, and analyze healthcare data. It empowers traditional medical devices with internet connectivity, enabling remote monitoring and real-time data analysis, which are pivotal in the realm of digital healthcare. These devices and applications establish connectivity through diverse means, including Wi-Fi, mobile networks, and mesh networks (D2D), ensuring seamless and efficient information exchange.

Recently, IoMT has seen significant advances in the deployment of wearables with Device-to-Device (D2D) communication for disaster management. These developments have revolutionized how healthcare professionals and emergency responders handle crisis situations. IoMT wearables equipped with D2D capabilities offer enhanced connectivity, allowing devices to communicate directly with one another, even in the absence of a centralized network, ensuring seamless data exchange in challenging environments. This feature is crucial during emergencies when traditional communication infrastructure may be compromised or overwhelmed. The integration of wearables with IoMT and D2D communication enables remote patient monitoring, providing real-time tracking of vital signs and physiological parameters of disaster-affected individuals. Such data empower rapid triage and decision-making, ensuring timely and appropriate medical interventions. Additionally, IoMT wearables with D2D facilitate predictive analytics and early warning systems, continuously monitoring physiological data to detect early signs of distress or illness in disaster-affected populations, enabling pre-emptive actions to mitigate health risks. Moreover, the generated data from these wearables provides valuable insights into disaster-affected areas, identifying hotspots, assessing injuries' severity, and optimizing resource allocation for efficient disaster response efforts. D2D-enabled IoMT wearables also serve as communication tools for first responders, enhancing coordination and information sharing among rescue teams. Furthermore, wearables equipped with biosensors can detect hazardous environmental factors in disaster zones, assisting in risk assessment and ensuring responder and survivor safety. These recent advances in IoMT and D2D have transformed disaster management, thereby providing real-time data insights and enabling more efficient and effective responses, ultimately saving lives and improving healthcare delivery during crises. The ongoing evolution of these technologies promises further transformative innovations for disaster management in the future ([39,40]).

The IoMT capability is integrated with the mPOC algorithm's second stage, where the system continuously scans and senses the user's physiological signals in an iterative loop. These signals are subject to meticulous analysis to calculate the user's Deterioration Risk Index (DRI) and Predictive Exposure Index (PEI) values based on their physiological data. Additionally, the system, which encompasses smartphones and/or Bio-Watches, proficiently captures COVID-19-related Clinical Presentation Profiles (CPPs), such as dry coughs, nasal congestion, and shortness of breath. By concurrently considering these CPPs alongside the PEI and DRI values, which fall within a range of 1 to 10, the system provides invaluable insights into the user's health condition.

6.3. Pandemic Response as a Service (PRaaS) Stage

During the final stage of the mPOC system, it establishes real-time bidirectional connections with other smartphones within the D2D wireless range (within 500 m), as well as with Public Health Broadcast Agencies (PHBA) and the Outbreak Tracking Center (OTC) through various network channels, such as mobile, internet, and cloud. Through these connections, the mPOC system collects anonymous location data from users within the wireless range and exchanges relevant information with the OTC. The PEI value, which reflects the environmental risk based on the user's location, activity type, and movement direction, and the DRI value, indicating the physiological risk factors if exposed to the virus, play a crucial role in this stage.

The concept of Pandemic Response as a Service (PRaaS) is rooted in the utilization of cloud technology to provide an effective pandemic response. First introduced in 2014 within the framework of Healthcare as a Service (HaaS) [41], PRaaS shares several common features with HaaS, including scalability based on regional and hierarchical structures,

seamless and on-demand operations, virtually infinite computation through fog and edge computing, big data analytics, online storage, and resource sharing [42,43]. However, it is important to highlight that the key differentiating factor of PRaaS lies in the integration of end-user smartphones equipped with Bio-Watch and running the mPOC platform, which form the backbone of the system's functionality.

6.4. DRI Construct

This subsection focuses on the calculation of DRI, and as illustrated in Table 2, it outlines the risk factors involved in determining the DRI value, including age, gender, ethnicity, height/weight, and pre-existing conditions. The relationships among these risk factors are as follows:

- 1. Age: Age plays a significant role in the outcomes following viral exposure. Generally, a higher age is associated with increased chances of complications. However, it is important to note that the severity of complications is also influenced by pre-existing conditions, which have notably contributed to higher mortality rates among younger patients.
- 2. Body Mass Index (BMI): BMI is calculated using weight (W) and height (H) with the formula W/H^2 (when weight is in kilograms and height is in meters), or BMI = $703 \times W/H^2$ (when weight is in pounds and height is in inches). Individuals who are classified as obese (BMI > 30) face a higher risk of complications during hospitalization, and this factor is included in the calculation of DRI.
- 3. Ethnicity: The mortality rate from COVID-19 has an association with patients' ethnic backgrounds, which is largely influenced by factors such as healthcare access, education, income, and social standing. These factors are taken into consideration when calculating DRI.
- 4. Pre-existing conditions: Based on reports from the World Health Organization (WHO), the increase in mortality rates for individuals with pre-existing health conditions (in addition to the rates for otherwise healthy individuals) are as follows: cardiovascular diseases (10.5%), diabetes (7.3%), pulmonary diseases (6.3%), hypertension (6%), and cancer (5.6%) [44]. These conditions are factored into the calculation of DRI.

The DRI value is presented as a color-coded indicator on the user's device, as mentioned in Table 3. This value is influenced by various scenarios, including the user's location at home and subsequent movement to other places. Therefore, the calculation of DRI is represented by Equation (1), indicating that the future (recursive) value of DRI is a function of the user's age, gender, BMI value, ethnicity, pre-existing medical conditions, and the current value of DRI.

$$DRI[i+1] = F(age, g, BMI, e, phc, DRI[i])$$
(1)

g: gender, e: ethnicity, phc: pre-existing health conditions.

Location (Now)	Location (Next)	PEI (Now)	DRI	PEI (Next)	Alert (Light-Audio)
Home	Home	\downarrow	\downarrow	\approx	G/S
Home	Home	\uparrow	\downarrow	\approx	Y/S
Home	Home	\downarrow	\uparrow	\approx	G/S
Home	Home	†	\uparrow	$\uparrow\uparrow$	A/S
Home	LRL	\downarrow	\downarrow	\approx	G/LRA
Home	LRL	†	\downarrow	$\uparrow\uparrow$	Y/LRA
Home	LRL	\downarrow	\uparrow	\approx	Y/LRA

 Table 3. User profile risk assessment criteria.

Location (Now)	Location (Next)	PEI (Now)	DRI	PEI (Next)	Alert (Light-Audio)
Home	LRL	\uparrow	\uparrow	\approx	A/LRA
Home	MRL	\downarrow	\downarrow	\approx	A/MRA
Home	MRL	\uparrow	\downarrow	$\uparrow \uparrow$	A/MRA
Home	MRL	\downarrow	\uparrow	\approx	R/MRA
Home	MRL	1	\uparrow	$\uparrow\uparrow$	R/MRA
Home	HRL	\downarrow	\downarrow	~	A/HRA
Home	HRL	\uparrow	\downarrow	$\uparrow \uparrow$	A/HRA
Home	HRL	\downarrow	1	*	R/HRA
Home	HRL	\uparrow	\uparrow	$\uparrow \uparrow$	R/HRA
MR	LRL	\uparrow	\downarrow	$\uparrow\uparrow\uparrow$	R/VHRA
HR	LRL	\uparrow	\downarrow	$\uparrow \uparrow \uparrow$	R/VHRA

Table 3. Cont.

LRL: Low-Risk Location; MRL: Medium-Risk Location; HRL: High-Risk Location; \downarrow : Low (1–5); \uparrow : High (5–10); $\uparrow\uparrow$: Higher (e.g., 3 to 8); $\uparrow\uparrow\uparrow$: Highest (e.g., 1 to 10); \approx : Remain; DRI Color Codes: Green (G), Yellow (Y), Amber (A), Red (R); PEI: Audio Codes: Silent (S), Low Risk Alert (LRA), Medium Risk Alert (MRA), High Risk Alert (HRA), Very High Risk Alert (VHRA).

6.5. PEI Construct

The Predictive Exposure Index (PEI) is also a recursive index primarily influenced by environmental and behavioral factors. In the initial stage, the algorithm collects information from the user regarding their household members, occupations, and other factors that impact their typical behaviors, as depicted in Equation (2).

During the IoMT stage, the algorithm monitors the user's physiological changes and captures environmental sounds, analyzing the rate at which they are changing, particularly in relation to DRI-related parameters.

In the PRaaS stage, PEI is actively updated by receiving location details from GPS and mobile networks, tracking the user's current location, movement, and intended final destination. The system regularly sends PEI updates to the Outbreak Tracking Center (OTC) and receives real-time updates from the Public Health Broadcasting Agencies and OTC. The DRI and PEI updates are translated into audio warnings for the users, indicating the risk level of their destination. Furthermore, users at the destination are informed about the risk rating of impending arrivals, showcasing the predictive nature of the system. The calculation of future PEI values involves considering both current DRI and PEI values in a recursive manner, as outlined in Equation (2). These dependencies, along with user behavior and movements, are presented in Table 3. As mentioned, the PEI/DRI calculation results are converted into audio and video signals, which users receive as health and location risk alerts and notifications (Table 3).

Upon initializing the mPOC application, PEI starts with a user-specific initial value. Based on Equation (2) and the user's movements, PEI is updated with new values. For instance, if the user remains stationary at home, the PEI value will remain low, and this information will be conveyed to the OTC. As the user ventures out of their residence, the PEI value may change depending on the risk factors associated with the trajectory or destination (for instance, going to a high-risk area will increase the PEI value). The DRI also continuously senses physiological parameters and updates its values accordingly. These two indices trigger appropriate light and audio notifications on the smartphone running the mPOC application to alert the user accordingly. Figure 3 shows the flowchart of the system and the three stages.

$$PEI[i + 1] = F(DRI[i], hh, OTC, lm, PEI[i])$$

$$(2)$$

15 of 25

hh: household data, *lm*: location and movement.

6.6. The mPOC System Methodology

The mPOC algorithm uses an iterative and recursive approach to provide continuous monitoring and adaptive capability to support specific users. Below is a detailed description of each stage aligned with the system's methodology and algorithm.

- (1) Stage 1 (Initial Stage):
 - Devices: Smartphone and Bio-Watch
 - Sensing/Information Input: The smartphone collects location data (suburb, city, country), BMI, pre-existing conditions, job information, and the number of members in the household. The Bio-Watch receives sensory data input from the smartphone and calibrates sensors to learn user's physiology.
 - Monitored CPPs/Health Conditions and User Inputs: Users input their age, gender, ethnicity, height, weight, and pre-existing conditions, including cancer, organ implants, hypertension, diabetes, and pulmonary, renal, CS, and cardiovascular diseases. The algorithm calculates BMI and risk factors based on age and gender, resulting in the calculation of DRI.
- (2) Stage 2 (IoMT Stage):
 - Devices: Smartphone and Bio-Watch
 - Sensing/Information Input: The smartphone gathers data from various sensors, such as microphone (voice/sound), camera (video/light), gyroscope, accelerometer, compass, and GPS. The Bio-Watch collects data from sensors, including temperature, accelerometer, gyroscope, compass, ECG, EMG, EEG, SpO2, BVP (Blood Velocity Pulse), SCL (Skin Conductance Level), and THz (Terahertz).
 - Monitored CPPs/Health Conditions and User Inputs: Users provide information about dry cough and nasal congestion through the microphone; shortness of breath through the microphone and SpO2 reading; hypertension through BVP sensor; pain level through BVP, ECG, and SCL sensors; fever through the temperature sensor; and tachycardia through ECG. Various diseases can also be monitored, such as restlessness/fatigue/insomnia using PSG (Polysomnography), confusion using EEG, diabetes using THz, heart conditions using ECG, and diarrhea and vomiting using the microphone.
 - Data Exchange: All the collected data is sent to the PRaaS stage, where the PEI is calculated.
- (3) Stage 3 (PRaaS Stage):
 - Devices: Smartphone and Bio-Watch
 - Sensing/Information Input: The smartphone uses LTE-Direct (D2D), connects with PHBA (Public Health Broadcasting Agency), and exchanges anonymous health details with users within wireless range. The PEI value is calculated and DRI and PEI values are recursively updated and sent to the OTC.

6.7. Simulation Environment and Parameter Setup

The mPOC system has been tested and simulated using the LabVIEW Community Edition [45], which is a powerful and versatile platform for data acquisition, analysis, and control. This subsection outlines the simulation environment setup and the related parameter configuration within the LabVIEW system.

(1) LabVIEW Simulation Environment: The LabVIEW Community Edition provides an ideal simulation environment for testing the mPOC system's functionalities in a controlled and repeatable manner. The LabVIEW graphical programming interface allows for easy integration of various components, making it suitable for simulating the complex interactions between wearable IoMT device emulation, data processing algorithms, and outbreak prediction models.

- (2) Wearable IoMT Device Emulation: To mimic real-world data acquisition from wearable IoMT devices, LabVIEW incorporates virtual instrumentation for emulating physiological signals, including EMG, EEG, ECG, and breathing patterns. This emulation allows for comprehensive testing of the mPOC system's data collection capabilities under different outbreak scenarios.
- (3) Data Processing Algorithms: LabVIEW's extensive library of built-in functions and tools is leveraged to develop and integrate data processing algorithms into the simulation environment. These algorithms simulate the real-time processing and analysis of user-centric physiological parameters, such as calculating Predictive Exposure Index (PEI) and Deterioration Risk Index (DRI) based on EMG, EEG, ECG, and breathing pattern data.
- (4) Outbreak Prediction Models: Within LabVIEW, sophisticated machine learning and artificial intelligence (AI) models are incorporated to predict outbreak trends and patterns based on the processed physiological data. These models take into account both bottom-up user-centric data and top-down public health information to make accurate outbreak predictions.
- (5) Parameter Setup: The LabVIEW simulation environment allows for flexible parameter configuration to test various outbreak scenarios and system responses. Key parameters include:
 - Outbreak Scenarios—Different outbreak scenarios—including varying disease transmission rates and geographic spread—can be simulated to assess the mPOC system's predictive capabilities under diverse conditions.
 - Physiological Signal Variability—LabVIEW allows for the introduction of signal noise and variability to emulate real-world fluctuations in physiological signals, enabling robustness testing of the system.
 - Network Latency—Parameters related to network connectivity and latency are adjustable to evaluate the system's performance in handling real-time data transmission from wearable IoMT devices to the central system.
 - Predictive Thresholds—The LabVIEW simulation environment facilitates the dynamic adjustment of predictive thresholds for PEI and DRI, enabling finetuning of the system's sensitivity and specificity in issuing outbreak warnings.

The accuracy and responsiveness of the mPOC system during various scenarios were tested, which are presented in Figure 4 and explained in detail in Section 7.2.



Figure 4. The performance measure (accuracy versus movement) of the mPOC system.

7. Feasibility, Scalability, and Performance Evaluation

As discussed in the previous sections, the mPOC system relies on a pair of devices, namely a smartphone and a Bio-Watch, with the mPOC application running on the smartphone. The system operates through three distinct modes of information data handling:

- (1) Capturing Clinical and Physiological Presentations (CPPs) data: This mode involves collecting relevant clinical and physiological information. The Bio-Watch and smartphone are linked using a Wireless Body Area Network (WBAN), typically utilizing technologies such as Bluetooth.
- (2) Bidirectional communication with the Outbreak Tracking Center (OTC): In this mode, the mPOC system gathers public health data and maintains continuous communication with the OTC, ensuring timely updates on outbreaks and relevant information.
- (3) Establishing Device-to-Device (D2D) links with other users within wireless range: The third mode enables the mPOC system to establish D2D connections with nearby smartphones, facilitating the exchange of vital details. These connections rely on widely available mobile networks, such as 4G LTE/5G, which are both scalable and capable of handling the necessary bandwidth. The mPOC system primarily transmits short text-based messages, resulting in low bandwidth requirements and eliminating the need for multimedia data exchange.

The computational analysis for the system can be easily performed on the smartphone itself, and the classification and characterization of health information can be accomplished using fog/edge computing techniques.

For DRI-related evaluations of the performance of the mPOC system in relation to COVID-19, reference [46] considers six clusters that indicate various infection complications. These clusters are as follows:

Cluster 1—flu-like without fever: Symptoms include headache, muscle and chest pains, dry cough, and sore throat.

Cluster 2—flu-like with fever: Similar to Cluster 1, but with the addition of fever and voice hoarseness.

Cluster 3—gastrointestinal: Symptoms consist of headache and chest pain, loss of smell and appetite, diarrhea, and sore throat without cough.

Cluster 4—severity 1, fatigue: This cluster includes headache, fatigue, loss of smell, dry cough, fever, voice hoarseness, and chest pain.

Cluster 5—severity 2, confusion: Symptoms comprise headache, loss of smell and appetite, dry cough, fever, voice hoarseness, sore throat, confusion, fatigue, and chest/muscle pain.

Cluster 6—severity 3, abdominal/respiratory: Similar to Cluster 5, but with the addition of shortness of breath, diarrhea, and abdominal pain.

These clusters indicate an increased likelihood of patients requiring supplemental oxygen or ventilators, ranging from 1.5% (Cluster 1) to 19.8% (Cluster 6). It's worth noting that most of these CPPs, except for loss of smell and appetite (which is less prevalent in the recent COVID-19 subaviants), are applicable to our mPOC system with a relatively high degree of accuracy (gold standard of 90% and above). Tracking the loss of smell/taste can be incorporated through a simple question-and-answer query issued to the user by the mPOC application.

7.1. The mPOC Technical Feasibility Study

The mPOC technical feasibility study evaluates various factors to determine if the system is feasible from technical, economical, operational, social, legal, ethnical, scalability, sustainability, and risk perspectives. In this article, we will only cover the technical feasibility aspect of the mPOC system. The feasibility of other factors can be considered in future studies.

The technical feasibility of the mPOC system focuses on evaluating the practicality and viability of its technology components. This assessment considers various technical aspects to determine if the system can be successfully developed, deployed, and operated. Here are the key factors involved in the technical feasibility of the mPOC system:

System Availability: This includes assessing the availability of hardware components, such as smartphones and wearable devices (e.g., Bio-Watch). It also considers the availability of software components, including the mPOC application and algorithms necessary for data processing and analysis.

Sensor Integration: The mPOC system relies on the integration of sensors to capture clinical and physiological data. This assesses the availability and functionality of sensors that can accurately and reliably measure vital signs, symptoms, and other relevant health data. This includes evaluating the compatibility of sensors with the smartphone and wearable device, ensuring seamless data collection and transmission.

Connectivity: The mPOC system requires connectivity options to transmit data between the smartphone, wearable device, and other public domain stakeholders, such as the OTC. This feasibility study examines the availability of wireless communication technologies, such as Bluetooth, Wi-Fi, or cellular networks (4G/5G), to establish reliable and secure connections. It ensures that the chosen connectivity options are capable of handling the required timely (real-time) information exchange.

Data Processing and Analysis: This required the computational evaluation of the capabilities of the smartphone to handle the processing and analysis of data collected by the mPOC system. This includes assessing the processing power, memory, and storage capacity of the smartphone to perform complex algorithms and classification tasks. Additionally, it considers the availability of fog/edge computing techniques to offload some computational tasks and improve efficiency.

Data Security and Privacy: This technical feasibility encompasses the assurance of robust (personal/public) data security and privacy measures within the mPOC system. This involves assessing encryption mechanisms, secure data transmission protocols, and access to control measures to protect sensitive health information. The system may require compliance with relevant data protection regulations and standards (e.g., NIST).

System Scalability: This examines the scalability of the mPOC system to accommodate a potentially large number of users. It assesses whether the system can handle increased data volume, simultaneous connections, and user interactions without compromising performance or data accuracy. Scalability considerations may involve evaluating the network infrastructure, server capabilities, and data storage capacity.

Integration with Existing Infrastructure: In many cases, the mPOC system needs to integrate with existing healthcare infrastructure and information systems. This feasibility study evaluates the compatibility and interoperability of the mPOC system with Electronic Health Record (EHR) systems, laboratory information systems, or public health databases. It ensures seamless integration and data exchange between the mPOC system and other healthcare systems.

By conducting a thorough Technical Feasibility assessment, the end-users can determine if the mPOC system's technology components are readily available, compatible, and capable of meeting the system's requirements. This evaluation provides valuable insights into its development.

7.2. The mPOC Performance Evaluation

In this section, the performance evaluation of the mPOC prediction model is analyzed. The mechanism involves capturing physiological data from users, which can be susceptible to unwanted noise and occasional faulty readings. To enhance the specificity and sensitivity of the diagnosis, appropriate filtering techniques, such as Kalman filtering, are employed [47].

The second mode of operation relies on the timely delivery of real-time health and outbreak information by the authority, necessitating the establishment and maintenance of anytime and anywhere connectivity. This is generally not a challenge in metropolitan areas. The third mode of operation involves establishing proper device-to-device (D2D) connections with community members for information exchange. Since no multimedia data, such as voice, video, or images, are transmitted across the network, there is no need to maintain a high-quality communication link. The transmitted information is in the form of text, allowing for some degree of delay.

The mPOC system is capable of tracking the physiological signs of users during the advanced stage (AS) if they require the attention of a healthcare professional while self-isolating at home.

For the end-to-end performance evaluation, we utilized LabVIEW Community Edition [45], simulating the mobile/Bio-Watch user, the Outbreak Tracking Center (OTC), and various infection clusters. The algorithm (Figure 3) is deployed and simulated revolving around the movement of the mobile user from a low-risk area (residence) to various highrisk areas. The local visual and auditory feedback systems were utilized and recorded during the movements. Figure 4 illustrates the achieved accuracy based on the movements from a Low-Risk Location (LRL) to a High-Risk Location (HRL) and vice versa.

The performance scatterplot demonstrates that the accuracy measures for correctly identifying the user's risk when moving from one LRL to another LRL are at the highest levels, with a mean value of 92.8% accuracy with the variations ranging from 91% to 95%. The accuracy measures slightly decrease for movements from LRL to HRL, with a mean value of 87.08% accuracy with the variations between 85% and 90%. The accuracy measures further decline for movements from HRL to HRL, with a mean value of 84.84% with the variations between 83% and 88%. This decline can be attributed to the increased utilization of the algorithm to capture, transmit, and interpret the rapidly changing parameters.

The accuracy measures display high variation during movements from HRL to LRL, with a mean value of 84.12% and significant variations between 79% and 89%. This variation is influenced by differences in parameter reporting between high- and low-profile locations.

A crucial aspect of the application that needs to be addressed is the security and privacy capabilities of the mPOC system. Since mPOC users are within the D2D wireless range of each other (within 500 m), they will receive personalized DRI and PEI values from other users, which are to be passed to the OTC. To address security and privacy requirements, each mPOC system is assigned a unique anonymous 10-digit code, which remains private and secured from all users. All communication channels between mPOC users are encrypted to prevent Man-In-The-Middle Attacks, ensuring that adversaries cannot impersonate valid users or hijack established communication channels. The OTC stores and maintains the list of validated secure codes for all users in its database.

Although the accuracy versus movement performance measure is by far the most important performance measure that can be considered in this article, other metrics can also be used to study the performance of the mPOC system, such as:

- (1) The economical, operational, social, legal, ethnical, scalability, and sustainability aspects of the mPOC system;
- The feasibility and readiness of the elderly and non-tech savvy users to use the mPOC technology for disaster management;
- (3) Delay studies for OTC to reach end-users;
- (4) The effectiveness of the mPOC system in keeping infected users in isolation once signs and symptoms of COVID-19 appear in a specific user, after which the system is capable of issuing appropriate warnings and instructions, thereby guiding the user to get tested at the nearest testing center and self-isolate at home.

Other performance measures may include the system's capability in contact tracing, which may also be possible and is envisioned in the system, which can be included in future studies.

8. Future Directions for Technology-Driven Pandemic Responses

Utilizing technology for containing pandemics and outbreaks, such as COVID-19, has demonstrated promising results and is crucial for achieving a safer community. As we look towards the future, it is evident that investing in the current technologies is the most viable

path forward. The following are important technology-driven capacities that should be developed and enhanced ([48–53]):

8.1. Leveraging Digital Technologies for Detection

Rapid viral detection is essential for containing viral pandemics with an R_0 higher than 1, which is the case for most diseases, including the flu and COVID-19. Various digital technologies can be utilized to increase the speed of disease detection, particularly when limited or no symptoms are present; the mPOC system serves as a perfect example of such a technology.

8.2. Real-Time Monitoring and Contact Tracing

Once an infection is detected and confirmed, it becomes crucial to monitor the infected population in real-time and perform contact tracing. Technologies that enable real-time monitoring and contact tracing play a vital role in managing the spread of the virus. However, because of the inherent delay in the operations of the conventional contact tracing systems and the rapid evolvements of the newer COVID-19 subvariants, which caused the R_0 values to increase by up to six-fold, the performance of the contact tracing systems gradually decreased from 67.1% to 15% [54]. The future technologies for contact tracing need to consider high R_0 values to ensure adequate performance. The mPOC system, when deployed at a population level, has the capability to fulfill this requirement.

8.3. Critical Data Sharing Capability

With the frequent movement of individuals across cities, states, and countries, the sharing of critical data becomes increasingly important. This data sharing helps in understanding and mitigating the emergence of new and more dangerous viral subvariants. Additionally, addressing the issue of social media disinformation and managing conflicting data are essential, and proper data assessment and sharing can contribute to tackling this challenge effectively.

8.4. Pandemic-Ready Smart Cities

A crucial aspect of smart cities is having a technology-enabled and interconnected city infrastructure. During a pandemic, having a predictive system that can show the trajectory of localized infections' severity becomes vital. By integrating pandemic readiness into smart city frameworks, cities can proactively respond to outbreaks and effectively allocate resources.

8.5. Top-Down Meeting Bottom-Up Approaches

The top-down approach involves information flowing from infrastructure, government agencies, and regulators down to end-users and the general population. Conversely, the bottom-up approach prioritizes the well-being of the general population. To achieve the best possible outcomes, both approaches should run in parallel. This approach, where top-down and bottom-up approaches converge, is adopted in the mPOC system and should be embraced in future technology-driven pandemic responses.

8.6. Vendor-Specific Bottom-Up Systems/Technologies

There are several recent gadgets and vendor-specific technologies that can be used for a bottom-up approach in a variety of pandemic-like situations. Here are a few examples:

- (1) Oura Ring: The Oura Ring is a wearable device that tracks various biometric data, including body temperature, heart rate, respiratory rate, and sleep patterns. It can provide early indications of changes in health and help individuals monitor their well-being [55].
- (2) BioButton: The BioButton is a wearable biosensor that continuously monitors vital signs, including temperature, heart rate, and respiratory rate. It provides real-time

health data and can help individuals identify potential COVID-19 symptoms and take appropriate actions [56].

Other vendor-specific technologies include: Garmin Vivosmart, Fitbit Sense, Apple Watch, and Samsung Galaxy Watch.

These are just a few examples of recent vendor-specific technologies that can be used for a bottom-up approach in a pandemic situation. It's important to note that the availability and specific features of these technologies may vary depending on the region and the vendors' offerings.

By focusing on the current state of gadgets, their future directions, and continuously advancing technology-driven approaches, we can enhance our ability to detect and contain pandemics effectively. By embracing digital technologies, real-time monitoring, data sharing, pandemic-ready smart cities, and integrating top-down and bottom-up approaches, we can build systems that can contribute to building resilient and proactive systems that can save lives and minimize the impact of future outbreaks.

9. Conclusions

In this paper, we have introduced the Mobile Health (mHealth) Predictive Outbreak COVID-19 (mPOC) system, which utilizes a cutting-edge wearable device that can be utilized by individuals from all walks of life. The centerpiece of this system is a smartphone-driven physiological watch called Bio-Watch, which captures and tracks users' physiological signals using both bottom-up (individuals first) and top-down—from public health and the Outbreak Tracking Center (OTC)—approaches.

By merging the top-down and bottom-up approaches, we believe we have found a way to effectively address fast evolving future pandemics and provide fast and efficient support to affected individuals. The algorithm described in this paper, which operates through three recursive stages, calculates the user's likelihood of being infected based on two formulas: the Predictive Exposure Index (PEI) and the Deterioration Risk Index (DRI). Visual and auditory warnings are then issued to inform users of their risk of infection.

We have also discussed the feasibility of this approach, aiming to revolutionize pandemic outbreak management by utilizing existing systems and sensors with minor upgrades to devices and communication protocols. The mPOC system makes optimal use of Device-to-Device (D2D) communication, which is supported by current 4G/5G networks. From a performance perspective, we simulated the behavior of the mPOC system using the LabView Community Edition. The results demonstrate mean accuracy values in correctly informing mobile users of risks, ranging from 92.8% for movements between Low-Risk Locations (LRL), 87.08% for movements from LRL to High-Risk Locations (HRL), 84.84% for movements within HRL, and 84.12% for movements from HRL to LRL. The decline in accuracies for movements within HRL can be attributed to the increased utilization of the algorithm to capture, transmit, and interpret parameters in high-risk environments.

The distinctive feature of the proposed method is its focus on the well-being of affected individuals and the ability to monitor them with accuracies close to the gold standard (>90%), which was previously unattainable. Our mPOC system has the potential to revolutionize pandemic management by providing timely and accurate information to individuals as well as public health agencies, thereby enabling early detection and intervention.

Further research and development in this field will allow us to refine the mPOC system and explore additional functionalities, such as contact tracing. By continually improving and advancing wearable IoMT technology, we can enhance the efficiency and effectiveness of outbreak prediction and monitoring, ultimately saving lives and mitigating the impact of future pandemics.

Funding: This research received no external funding.

Data Availability Statement: Not applicable.

22 of 25

Conflicts of Interest: The author declares no conflict of interest.

Abbreviations

3GPP	3rd Generation Partnership Project
4G/5G/6G	4th/5th/6th Generation Mobile Network
А	Amber (color code)
AS	Advanced Stage
Bio-Watch	Biological Watch
BMI	Body Mass Index
BVP	Blood Volume Pulse
CPP	Clinical and Physiological Presentation
CS	Cytokine Storm
D2D	Device-to-Device
DRI	Deterioration Risk Index
ECG	Electrocardiography
EEG	Electroencephalography
EHR	Electronic Health Record
FMG	Flectromyography
FS	Farly Stage
FDA	Food and Drug Administration
G	Green (color code)
н	Height
HaaS	Healthcare as a Service
HRA	High-Risk Alert (audio Codes)
HRI	High-Risk Location
Inte	Internet of Medical Things
IP	Incubation Period
	Low-Risk Alert (audio Codes)
IRI	Low Risk Location
LKL	Long Term Evolution
	Long Term Evolution
mDPS	m Hoalth Disaster Pacework System
MEDC	Middle East Beggingtony System
MEK5	Mabile Health
mPOC	m Haalth Dradictive Outbreak for COVID 10
MDA	Madium Pick Alart (audia Cadas)
MDI	Medium-Risk Alert (audio Codes)
MKL	Medium-Kisk Location
NFC	Near Field Communication
NISI OTC	National Institute of Standards and Technology
DIC	
PCR	Polymerase Chain Reaction
PEI	Predictive Exposure Index
PGS	Polysomnography
PHBA	Public Health Broadcast Agencies
PRaaS	Pandemic Response As A Service
R	Red (color code)
RAT	Rapid Antigen Test
5	Silent (audio codes)
SARS	Severe Acute Respiratory Syndrome
SCL	Skin Conductance Level
SpO2	Oxygen Saturation
SS	Specificity and Sensitivity
THz	Terahertz Technology
VHR	Very High-Risk Alert (audio code)
W	Weight

WBAN	Wireless Body Area Network
WHO	World Health Organization
Y	Yellow (color code)

References

- 1. Coronavirus Statistics. Worldometer. Available online: https://www.worldometers.info/coronavirus (accessed on 26 June 2023).
- 2. Wong, E.; Ho, K.F.; Wong, S.Y.; Cheung, A.W.; Yeoh, E. Workplace safety and coronavirus disease (COVID-19) pandemic: Survey of employees. *Bull. World Health Organ.* **2020**, *98*, 150. [CrossRef]
- 3. World Health Organization. Transmission of SARS-CoV-2: Implications for Infection Prevention Precautions: Scientific Brief. 2020. Available online: https://www.who.int/news-room/commentaries/detail/transmission-of-sars-cov-2-implications-for-infection-prevention-precautions (accessed on 12 April 2022).
- 4. Adibi, S.; Rajabifard, A.; Islam, S.; Ahmadvand, A. COVID-19 Pandemic: Lessons Learnt and Roadmap for the Future. In *The Science behind the COVID Pandemic and Healthcare Technology Solutions*; Springer Series in Bio-/Neurosystems: Berlin/Heidelberg, Germany, 2022; ISBN 978-3-031-10030-7.
- Adibi, S. Mobile Health Personal-to-Wide Area Network Disaster Management Paradigm. IEEE Sens. J. 2018, 18, 9874–9881. [CrossRef]
- Pandit, J.A.; Radin, J.M.; Quer, G.; Topol, E.J. Smartphone apps in the COVID-19 pandemic. *Nat. Biotechnol.* 2022, 40, 1013–1022. [CrossRef]
- Schmeelk, S.; Davis, A.; Li, Q.; Shippey, C.; Utah, M.; Myers, A.; Turchioe, M.R.; Creber, R.M. Monitoring Symptoms of COVID-19: Review of Mobile Apps. *JMIR mHealth uHealth* 2022, 10, e36065. [CrossRef] [PubMed]
- 8. Farzandipour, M.; Nabovati, E.; Sharif, R. The effectiveness of tele-triage during the COVID-19 pandemic: A systematic review and narrative synthesis. *J. Telemed. Telecare* 2023, 1357633X221150278. [CrossRef]
- Kagiyama, N.; Hiki, M.; Matsue, Y.; Dohi, T.; Matsuzawa, W.; Daida, H.; Minamino, T.; Kasai, T. Validation of telemedicine-based self-assessment of vital signs for patients with COVID-19: A pilot study. J. Telemed. Telecare 2021, 1357633X211011825. [CrossRef] [PubMed]
- Alhajri, N.; Simsekler, M.C.E.; Alfalasi, B.; Alhashmi, M.; AlGhatrif, M.; Balalaa, N.; Al Ali, M.; Almaashari, R.; Al Memari, S.; Al Hosani, F.; et al. Physicians' Attitudes Toward Telemedicine Consultations During the COVID-19 Pandemic: Cross-sectional Study. JMIR Public Health Surveill. 2021, 9, e29251. [CrossRef] [PubMed]
- 11. Maroju, R.G.; Choudhari, S.G.; Shaikh, M.K.; Borkar, S.K.; Mendhe, H.; Maroju, R.G., Jr.; Borkar, S. Role of telemedicine and digital technology in public health in India: A narrative review. *Cureus* **2023**, *15*, e35986. [CrossRef]
- Park, H.S.; Jeong, S.; Chung, H.-Y.; Soh, J.Y.; Hyun, Y.H.; Bang, S.H.; Kim, H.S. Use of video-based telehealth services using a mobile app for workers in underserved areas during the COVID-19 pandemic: A prospective observational study. *Int. J. Med. Inform.* 2022, 166, 104844. [CrossRef]
- 13. Patel, P.; Kerzner, M.; Reed, J.B.; Sullivan, P.S.; El-Sadr, W.M. Public Health Implications of Adapting HIV Pre-exposure Prophylaxis Programs for Virtual Service Delivery in the Context of the COVID-19 Pandemic: Systematic Review. *JMIR Public Health Surveill.* 2022, *8*, e37479. [CrossRef]
- Breckner, A.; Litke, N.; Göbl, L.; Wiezorreck, L.; Miksch, A.; Szecsenyi, J.; Wensing, M.; Weis, A. Effects and Processes of an mHealth Inter-vention for the Management of Chronic Diseases: Prospective Observational Study. *JMIR Form. Res.* 2022, 6, e34786. [CrossRef] [PubMed]
- 15. Yuen, K.F.; Koh, L.Y.; Wong, Y.Q.; Wang, X. Sustainable crowdsourced delivery: A study of technological, health, value, and trust antecedents of consumer loyalty. J. Clean. Prod. 2023, 405, 137010. [CrossRef]
- 16. von Wyl, V.; Höglinger, M.; Sieber, C.; Kaufmann, M.; Moser, A.; Serra-Burriel, M.; Puhan, M.A. Drivers of acceptance of COVID-19 proximity tracing apps in Switzerland: Panel survey analysis. *JMIR Public Health Surveill.* **2021**, *7*, e25701.
- 17. Munzert, S.; Selb, P.; Gohdes, A.; Stoetzer, L.F.; Lowe, W. Tracking and promoting the usage of a COVID-19 contact tracing app. *Nat. Hum. Behav.* **2021**, *5*, 247–255. [CrossRef]
- 18. Ali, Y.; Khan, H.U. A Survey on harnessing the Applications of Mobile Computing in Healthcare during the COVID-19 Pandemic: Challenges and Solutions. *Comput. Netw.* **2023**, 224, 109605. [CrossRef] [PubMed]
- 19. Rosen, A. What You Need to Know About XBB.1.5, the Latest Omicron Variant, John Hopkins University. 2023. Available online: https://publichealth.jhu.edu/2023/what-you-need-to-know-about-xbb15-the-latest-omicron-variant (accessed on 5 May 2023).
- Karyakarte, R.P.; Das, R.; Rajmane, M.V.; Dudhate, S.; Agarasen, J.; Pillai, P.; Chandankhede, P.M.; Labhshetwar, R.S.; Gadiyal, Y.; Kulkarni, P.P.; et al. Chasing SARS-CoV-2 XBB.1.16 Recombinant Lineage in India and the Clinical Profile of XBB.1.16 Cases in Maharashtra, India. *Cureus* 2023, 15, e39816. [CrossRef] [PubMed]
- Foresight and Analysis of Infectious Disease Threats to Virginia's Public Health, University of Virginia, Biocomplexity Institute Technical Report: TR BI-2023-149. 2023. Available online: https://biocomplexity.virginia.edu/system/files/publications/2023-0 3/COVID-19_VA_UVA-BI-Update_02-March-2023%20%281%29.pdf (accessed on 25 April 2023).
- 22. Adibi, S.; Nguyen, L.; Hamper, A.; Bodendorf, F.; Wickramasinghe, N. A systematic approach to terahertz-based glucose monitoring. In Proceedings of the the 30th Bled eConference, Digital Transformation—From Connecting Things to Transforming Our Lives, Bled, Slovenia, 18–21 June 2017.

- 23. Report of the WHO-China Joint Mission on Coronavirus Disease 2019 (COVID-19). 2020. Available online: https://www.who. int/docs/default-source/coronaviruse/who-china-joint-mission-on-covid-19-final-report.pdf (accessed on 9 March 2020).
- Menni, C.; Valdes, A.M.; Polidori, L.; Antonelli, M.; Penamakuri, S.; Nogal, A.; Louca, P.; May, A.; Figueiredo, J.C.; Hu, C.; et al. Symptom prevalence, duration, and risk of hospital admission in individuals infected with SARS-CoV-2 during periods of omicron and delta variant dominance: A prospective observational study from the ZOE COVID Study. *Lancet* 2022, 399, 1618–1624. [CrossRef]
- Struyf, T.; Deeks, J.J.; Dinnes, J.; Takwoingi, Y.; Davenport, C.; Leeflang, M.M.; Spijker, R.; Hooft, L.; Emperador, D.; Dittrich, S.; et al. Signs and symptoms to determine if a patient presenting in primary care or hospital outpatient settings has COVID-19 disease. *Cochrane Database Syst. Rev.* 2020, 2020, CD013665. [CrossRef]
- Yin, Y.; Lin, J.; Yuan, S.; Tong, S.; Chen, E.; Zheng, J.; Wang, W. The relationship between early isolation and the duration of viral shedding of mild and asymptomatic infection with SARS-CoV-2 Omicron BA.2 variant. J. Infect. 2022, 85, e184–e186. [CrossRef]
- 27. Rowe, B.R.; Canosa, A.; Meslem, A.; Rowe, F. Increased airborne transmission of COVID-19 with new variants, implications for health policies. *Build. Environ.* 2022, 219, 109132. [CrossRef]
- 28. Velavan, T.P.; Meyer, C.G. The COVID-19 epidemic. Trop. Med. Int. Health 2020, 25, 278–280. [CrossRef]
- 29. Liu, Y.; Rocklöv, J. The effective reproductive number of the Omicron variant of SARS-CoV-2 is several times relative to Delta. *J. Travel Med.* 2022, *29*, taac037. [CrossRef] [PubMed]
- Liu, Y.; Rocklöv, J. The reproductive number of the Delta variant of SARS-CoV-2 is far higher compared to the ancestral SARS-CoV-2 virus. J. Travel Med. 2021, 28, taab124. [CrossRef] [PubMed]
- Sheikhi, F.; Yousefian, N.; Tehranipoor, P.; Kowsari, Z. Estimation of the basic reproduction number of Alpha and Delta variants of COVID-19 pandemic in Iran. *PLoS ONE* 2022, 17, e0265489. [CrossRef]
- 32. Coburn, B.J.; Wagner, B.G.; Blower, S. Modeling influenza epidemics and pandemics: Insights into the future of swine flu (H1N1). BMC Med. 2009, 7, 30. [CrossRef] [PubMed]
- Yi, W.J.; Saniie, J. Patient Centered Real-Time Mobile Health Monitoring System. *E-Health Telecommun. Syst. Netw.* 2016, 5, 75. [CrossRef]
- 34. Number of mobile 4G/LTE Subscriptions Worldwide by Region from 2011–2028. 2023. Available online: https://www.statista. com/statistics/521572/4g-5g-mobile-subscriptions-worldwide (accessed on 19 June 2023).
- Gismalla, M.S.M.; Azmi, A.I.; Bin Salim, M.R.; Abdullah, M.F.L.; Iqbal, F.; Mabrouk, W.A.; Othman, M.B.; Ashyap, A.Y.I.; Supa'At, A.S.M. Survey on Device to Device (D2D) Communication for 5GB/6G Networks: Concept, Applications, Challenges, and Future Directions. *IEEE Access* 2022, 10, 30792–30821. [CrossRef]
- Vidakis, K.; Mavrogiorgou, A.; Kiourtis, A.; Kyriazis, D. A comparative study of short-range wireless communication technologies for health information ex-change. In Proceedings of the 2020 International Conference on Electrical, Communication, and Computer Engineering (ICECCE), Istanbul, Turkey, 12–13 June 2020.
- Kiourtis, A.; Mavrogiorgou, A.; Kyriazis, D.; Graziani, A.; Torelli, F. Improving Health Information Exchange through Wireless Communication Protocols. In Proceedings of the 2020 16th International Conference on Wireless and Mobile Computing, Networking and Communications (WiMob), Thessaloniki, Greece, 12–14 October 2020; pp. 32–39. [CrossRef]
- Guest, P.C.; Abbasifard, M.; Jamialahmadi, T.; Majeed, M.; Kesharwani, P.; Sahebkar, A. Multiplex Immunoassay for Prediction of Disease Severity Associated with the Cytokine Storm in COVID-19 Cases. *Methods Mol. Biol.* 2022, 2511, 245–256. [CrossRef]
- Razdan, S.; Sharma, S. Internet of medical things (IoMT): Overview, emerging technologies, and case studies. *IETE Tech. Rev.* 2022, 39, 775–788. [CrossRef]
- 40. Vishnu, S.; Ramson, S.J.; Jegan, R. Internet of medical things (IoMT)-An overview. In Proceedings of the 2020 5th International Conference on Devices, Circuits and Systems (ICDCS), Coimbatore, India, 5–6 March 2020.
- John, N.; Shenoy, S. Health cloud—Healthcare as a service(HaaS). In Proceedings of the 2014 International Conference on Advances in Computing, Communications and Informatics (ICACCI), Delhi, India, 24–27 September 2014; pp. 1963–1966. [CrossRef]
- 42. Khadhraoui, M.; Bellaaj, H.; Ammar, M.B.; Hamam, H.; Jmaiel, M. Survey of BERT-base models for scientific text classi-fication: COVID-19 case study. *Appl. Sci.* **2022**, *12*, 2891. [CrossRef]
- 43. Aceto, G.; Persico, V.; Pescapé, A. Industry 4.0 and Health: Internet of Things, Big Data, and Cloud Computing for Healthcare 4.0. *J. Ind. Inf. Integr.* **2020**, *18*, 100129. [CrossRef]
- 44. Yaddanapudi, L.N. Comorbidities and COVID-19. J. Anaesthesiol. Clin. Pharmacol. 2020, 36, S18–S20. [CrossRef] [PubMed]
- LabVIEW Community (Ed.). Available online: https://www.ni.com/en-au/shop/labview/select-edition/labview-communityedition.html (accessed on 28 October 2022).
- Wise, J. COVID-19: Study reveals six clusters of symptoms that could be used as a clinical prediction tool. *BMJ* 2020, 370, m2911. [CrossRef]
- 47. Rao, V.C.S.; Devi, B.G.; Pratapagiri, S.; Srinivas, C.; Venkatramulu, S.; Raghavakumari, D. Prediction of COVID-19 using Kalman filter algorithm. *AIP Conf. Proc.* 2022, 2418, 030067. [CrossRef]
- 48. Kummitha, R.K.R. Smart technologies for fighting pandemics: The techno- and human- driven approaches in controlling the virus transmission. *Gov. Inf. Q.* 2020, *37*, 101481. [CrossRef] [PubMed]
- 49. Khanna, R.C.; Cicinelli, M.V.; Gilbert, S.S.; Honavar, S.G.; Murthy, G.S.V. COVID-19 pandemic: Lessons learned and future directions. *Indian J. Ophthalmol.* 2020, *68*, 703–710. [CrossRef]

- 50. Budd, J.; Miller, B.S.; Manning, E.M.; Lampos, V.; Zhuang, M.; Edelstein, M.; Rees, G.; Emery, V.C.; Stevens, M.M.; Keegan, N.; et al. Digital technologies in the public-health response to COVID-19. *Nat. Med.* **2020**, *26*, 1183–1192. [CrossRef]
- OECD. How will COVID-19 Reshape Science, Technology and Innovation? 2021. Available online: https://www.oecd.org/ coronavirus/policy-responses/how-will-covid-19-reshape-science-technology-and-innovation-2332334d (accessed on 5 September 2022).
- Six Ways UNICEF is Innovating to Respond to the Pandemic and Build Stronger Health Systems, OFFICE of Innovation, UNISEF. Available online: https://www.unicef.org/innovation/six-innovative-ways-pandemic-response-health-systems (accessed on 10 September 2022).
- Al Knawy, B.; McKillop, M.M.; Abduljawad, J.; Tarkoma, S.; Adil, M.; Schaper, L.; Chee, A.; Bates, D.W.; Klag, M.; Lee, U.; et al. Successfully Implementing Digital Health to Ensure Future Global Health Security during Pandemics. *JAMA Netw. Open* 2022, 5, e220214. [CrossRef]
- 54. Vogt, F.; Haire, B.; Selvey, L.; Katelaris, A.L.; Kaldor, J. Effectiveness evaluation of digital contact tracing for COVID-19 in New South Wales, Australia. *Lancet Public Health* **2022**, *7*, e250–e258. [CrossRef]
- 55. Winter, C.W. Oura Ring Gen3: We Tried It, and Here's Our Unfiltered Opinion, Men's Health. 2023. Available online: https://www.menshealth.com/fitness/a44110211/oura-ring-review (accessed on 26 June 2023).
- 56. BioIntelliSense BioButtonTM. Available online: https://www.biointellisense.com (accessed on 26 June 2023).

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.